Author's response to reviews

Title: A new combined strategy to implement a community occupational therapy intervention: designing a cluster randomized controlled trial

Authors:

Carola M.E. Döpp (cme_dopp@hotmail.com)
Maud J.L. Graff (M.Graff@iq.umcn.nl)
Steven Teerenstra (S.Teerenstra@ebh.umcn.nl)
Eddy Adang (E.Adang@ebh.umcn.nl)
Ria W.G. Nijhuis - van der Sanden (R.Nijhuis@iq.umcn.nl)
Marcel G.M. OldeRikkert (M.Olde-Rikkert@ger.umcn.nl)
Myrra J.F.J. Vernooij - Dassen (M.Vernooij-Dassen@iq.umcn.nl)

Version: 3 Date: 22 March 2011

Author's response to reviews: see over
Dear Mr / Ms,

As requested changes were made in the format of the manuscript. In this letter I would like to explain some of the changes or respond to some of the requests.

The following changes in the manuscript were made:

- The format and order of the names of the authors was revised to match the names entered into the BMC electronic submission system.
- Reference 7, 8, and 24 were removed from the reference list as requested and cited in text.
- I included the three tables in the main manuscript so they will be visible within the article.

Some of the minor changes were made as well. I would like to request BMC to make the changes that I did not address. The following changes were made by the author:

- The format of the title was changed.
- All affiliations now include details on the city and the country.
- The name of the section Methods was changed into Methods/Design in the abstract as well as in the main manuscript.

If additional information is required please contact me by email.

With kind regards,

Carola Döpp, MSc OT
Junior Researcher / PhD candidate
Scientific Institute for Quality of Healthcare (IQ healthcare)
Radboud University Nijmegen Medical Centre
P.O. Box 9101, 114 IQ healthcare
6500 HB Nijmegen
The Netherlands

T +31 24-3667310/ F +31 24 35 40 166
E cme_dopp@hotmail.com
I www.iqhealthcare.nl